Author's response to reviews

Title: Follow up of patients who start treatment with antidepressants: treatment satisfaction, treatment compliance, efficacy and safety

Authors:

Jesus Lopez-Torres Hidalgo (jesusl@sescam.org)
Ignacio Párraga Martínez (iparraga@sescam.jccm.es)
José María del Campo del Campo (jmariadc@sescam.jccm.es)
Alejandro Villena Ferrer (alejandro@sescam.jccm.es)

Version: 4 Date: 19 December 2012

Author's response to reviews:

Dear Deesha Majithia:

Following your instructions, I have resubmitted the manuscript without additional files. I would like to clarify that the study was not funded by my own institution. My institution is the Health Care Service of Castilla-La Mancha (SESCAM) and the institution which funded the study is the Foundation for Health Research of Castilla-La Mancha (FISCAM). The study obtained financiación by via competitive in a call for research grants.

If despite this, you think that the manuscript should be peer-reviewed, there is no objection on my part.

Best wishes